Author’s response to reviews

Title: Undergraduate medical education in general practice/family medicine throughout Europe - a descriptive study

Authors:

Mette Brekke (mette.brekke@medisin.uio.no)
Francesco Carelli (francesco.carelli@unimi.it)
Natalia Zarbailov (zarbailovnatalia@gmail.com)
Givi Javashvili (gjavashvili@tsmu.edu)
Stefan Wilm (stefan.wilm@med.uni-duesseldorf.de)
Markku Timonen (markku.timonen@oulu.fi)
Howard Tandeter (Howard1@bgu.ac.il)

Version: 2 Date: 6 October 2013

Author’s response to reviews:

The response to the reviewers is included in the cover letter.